Author’s response to reviews

Title: Estimating the case numbers of hand, foot and mouth disease among children under-five in Beijing during 2012, based on a telephone survey of healthcare-seeking behavior

Authors:

Xiaoli Wang (wangxiaoli198215@163.com)
Xiaona Wu (pinweibaobei@126.com)
Lei Jia (lailajia@126.com)
Xitai Li (lixitai0822@163.com)
Jie Li (lijie304@sina.com)
Shuang Li (li_shuang176@hotmail.com)
Haikun Qian (qiankun620@163.com)
Quanyi Wang (bjcdcxm@126.com)

Version: 3
Date: 6 August 2014

Author’s response to reviews: see over
Response to reviewer

Reviewer #1: Mark Riddle

Quality of written English: Not suitable for publication unless extensively edited

Author: We agree with the reviewer’s comments about the quality of written English of this manuscript. We have had two English speaking colleagues- C Raina MacIntyre and Holly Seale to help us copyedit the paper. Changes can be seen throughout the manuscript.